Author’s response to reviews

Title: The Candida species that are important for the development of atrophic glossitis in xerostomia patients

Authors:

Sachika Nakamura (dens32@cyber.ocn.ne.jp)
Mariko Okamoto (okamari616@gmail.com)
Ken Yamamoto (yamamoto-k@tsurumi-u.ac.jp)
Akihisa Tsurumoto (tsurumoto-a@tsurumi-u.ac.jp)
Yoko Yoshino (yoshino_yoko@isc.sagami-wu.ac.jp)
Hiroshi Iwabuchi (iwabuchi@kdu.ac.jp)
Ichiro Saito (saito-i@tsurumi-u.ac.jp)
Nobuko Maeda (maeda-n@tsurumi-u.ac.jp)
Yoichi Nakagawa (nakagawa-y@tsurumi-u.ac.jp)

Version: 3 Date: 09 Sep 2017

Author’s response to reviews:

Thank you for your review of our paper. A point-by-point response has been provided below.

1. Please rename the "Materials and methods" section to "Methods" in order for the manuscript format to meet the journal guidelines.

Response: We have renamed. 10th line, Page 5.

2. Please include the sentence "All authors have read and approved the final version of this manuscript" in the Author contributions section and ensure that this is the case.

Response: We have included the sentence. 14-15th line, Page 15.
3. If no specific funding was received for this study, please state this.

Response: We have stated in the Funding. 7th line, Page 15.

4. Please upload a clean version of the manuscript without tracked changes or highlighting and then remove any highlighted versions from the inventory, as these are no longer required at this stage of the editorial process.